Author's response to reviews

Title: Chest Pain with ST segment elevation in a patient with prosthetic aortic valve infective endocarditis: a case report.

Authors:

Vishal Luther Dr (vishal_luther@yahoo.co.uk)
Refai Showkathali Dr (refais@gmail.com)
Reto Gamma Dr (reto.gamma@btuh.nhs.uk)

Version: 3 Date: 21 June 2011

Author's response to reviews:

Dear Editorial Team

Many thanks for your email

I have edited the manuscript to address the requests that you have made

1) include the patient's ethnicity in the abstract and case presentation
   “British-Caucasian”
2) reformat the title in the style "A presenting with B in C: a case report"

In addition, the manuscript itself has been revised a little to improve the quality of the manuscript.

Regards

Vishal Luther